Author's response to reviews

Title: Mid-term Results of Stryker Scorpio Plus Mobile Bearing Total Knee Arthroplasty

Authors:

Hideo Kobayashi (hi-deo@live.jp)
Naoto Mitsugi (nmitsugi@uraph.yokohama-cu.ac.jp)
Yuichi Mochida (U1Mochida@aol.com)
Naoya Taki (ntak@peach.plala.or.jp)
Yasushi Akamatsu (akamatsu@yokohama-cu.ac.jp)
Masato Aratake (aratake@cj9.so-net.ne.jp)
Hirohiko Ota (ota2000@sf6.so-net.ne.jp)
Katsushi Ishii (ishi69@yokohama-cu.ac.jp)
Kengo Harigane (harigane-ykh@umin.ac.jp)
Taichi Ideno (i-taichi@tulip.san-net.ne.jp)
Tomoyuki Saito (t_saito@med.yokohama-cu.ac.jp)

Version: 3 Date: 20 January 2012

Author's response to reviews: see over
January 20, 2012

Hideo Kobayashi, MD., Ph.D
Department of Orthopedic Surgery
Yokohama City University Medical Center
4-57 Urafune-cho, Minami, Yokohama
Kanagawa, Japan, 232-0024
Office: +81-45-261-5656
Fax: +81-45-252-7470
E-mail: hi-deo@live.jp

Editors
Sports Medicine Arthroscopy Rehabilitation Therapy Technology,
Department of Orthopaedic Surgery,
Kobe University Graduate School of Medicine,
7-5-1 Kusunoki- cho, Chuo- ku, Kobe 650-0017, Japan

Dear Drs. Masahiro Kurosaka and Kai- Ming Chan

Please find enclosed revised manuscript entitled: Mid- term Results of Stryker ® Scorpio Plus Mobile Bearing Total Knee Arthroplasty”. We appreciate the reviewers’ comments and have modified the manuscript as indicated below.

I confirm that all authors were fully involved in the study and preparation of the manuscript and that the material within has not been and will not be submitted for publication elsewhere.
All authors certify they have not signed any agreement with a commercial interest related to this study which would in any way limit publication of any and all data generated for the study or to delay publication for any reason.

We hope that our manuscript will now be suitable for publication.
Feel free to contact me if you have other questions or recommendations.

Sincerely yours,
Dear Editor.

This concerns our revised manuscript entitled; Mid- term Results of Stryker® Scorpio Plus Mobile Bearing Total Knee Arthroplasty.

We appreciate the reviewers comments and have modified the manuscript as indicated below. Changes are also highlighted in red in the text.

Reviewer Comments:

1. The revised manuscript will be fit to publication. However, I would suggest to delete "although there were no significant differences between two groups." from the conclusion of both abstract and text.

   Thank you very much for your comment.
   We agree with your comment. We have deleted the sentence from the conclusion of both abstract and text.

Sincerely,